0 0.2 0.4 0.6 0.8 1 1.2

10^{-9}

10^{-8}

10^{-7}

10^{-6}

10^{-5}

10^{-4}

-1 )^2 > (100 MeV/c ch >/<N ee /dm ee <dN

)^2

(GeV/c eem

CERES/NA45 Pb-Au 40 AGeV

Preliminary 30 %≈_{geo}σ/σ

>=210 η/d ch <dN <2.65 η 2.1<

>200 MeV/c tp >35 mrad eeΘ

γ ee →^0π γee →η

0π ee →ω

γee →^⁄η ee →ρ ee →ω

ee →ϕ

Number of pairs for m>0.2 GeV/c2: 180+-48 Ratio Signal/Background: 1/6

Hadronic decay cocktail: - particle ratios taken from thermal model for Pb-Pb - rapidity and pt distributions from systematics in Pb-Pb

Enhancement: measured pairs/decay cocktail: 5.0 +- 1.3 region) ω - folded with experimental resolution (4.2% in the